\begin{table}%t3
\caption {\label{barsize}Bar sizes.  The bar size derivation is described in Sect.~\ref{morphology}.}
%\centerline
 {\begin{tabular}{l l l }     % 14columns
\hline \hline
 Galaxy name & Bar size (arcsec) & Bar size (arcsec)  \\ 
        & (other studies)   & (our study)  \\
\hline
NGC~1169 &   --                &  29   \\
NGC~1358$^{d}$ &   25    &   --    \\
NGC~1433$^{b, i}$&   90, 66 &   --               \\
NGC~1530$^{e}$ &  50                 & 69  \\
NGC~1832$^{f}$& 17 &  --     \\
NGC~2217$^{b; f; i}$ &50, 37, 43  &   -- \\        
NGC~2273 $^{a}$& 14  &  21           \\     
NGC~2523 &   --       &  --    \\      
NGC~2665 &    --                &  --    \\    
NGC~2681$^{1; a; c; g}$& 50, 25, 29  &  23          \\       
NGC~2859$^{b; c}$ &34, 46  &  48            \\       
NGC~2935$^{f; i}$&25, 39 &--                          \\
NGC~2950 $^{a; g}$& 24, 38 &  44            \\       
NGC~2962 $^{a}$&  29 &  45            \\   
NGC~3081$^{b; g; c}$ &35, 41, 38  &  35            \\      
NGC~4245$^{a}$ &  37&  59          \\        
NGC~4314 $^{a, g}$& 67, 75  &  92    \\      
NGC~4394$^{h;i}$ & 52,  35    & 56\\      
NGC~4643$^{a}$ & 50   & 67    \\  
NGC~5101$^{f}$& 50 & --  \\ 
 \hline
\end{tabular}}
\par
\smallskip
 {
(1) NGC~2681 is a triple barred system. Erwin (\cite{Erwin05}) gives the value of the primary bar, while we give here the value of the secondary bar.\\
$^{a}$~Erwin   (\cite{Erwin05}) bar size given as the radius of maximum ellipticity in the bar region;
$^{b}$~Erwin (\cite{Erwin04}) bar size given at the radius of maximum ellipticity in the bar region;
$^{c}$~Wozniak (\cite{Wozniak95}); they give the projected bar size at the radius of minimum ellipticity;
$^{d}$~Mulchaey (\cite{Mulchaey97}); they give the bar size as the radius at the minimum of ellipticity;
$^{e}$~Regan (\cite{Regan96});
$^{f}$~Jungwiert (\cite{Jungwiert97}), bar size given as the radius of maximum ellipticity in the bar region;
$^{g}$~Friedli et~al. (\cite{Friedli96}); they give the projected bar size as the radius at the ellipticity minimum;
$^{h}$~Gadotti \& de Souza (\cite{Gadotti06}), bar size given as the radius of maximum ellipticity in the bar region;
$^{i}$ Martin \& Roy (\cite{Martin95}), bar size given by visual inspection of photographic plates.
}
\end{table}